
Daily status, October 16
10/16/13

Summary:
Fixed Rlw.out.far pyrgeometer once again.
Today will attempt to hardwire floor Zigbee motes (radiation and soil) to the DSM.

T/RH: RH.40m.near and RH.35m.rim perhaps a bit high.
P: ok
csat u,v: ok
csat ldiag: ok
csat w, tc: ok
kh2o: FLR variance high at night, but also tc'tc'
motes: FAR ok 
           NEAR ok
           FLR all off and on; outages generally last 1:00 - 1:05 hrs.
Wetness: ok
radiation: Rpile.out.far ok
Tsoil: Tsoil.2.5cm.flr & far (linear avg) does not fit profile
Gsoil: ok
Qsoil: ok
Cvsoil: ok
2D sonic: ok
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